

# OPEN ACCESS

**Citation:** Tymejczyk O, Brazier E, Yiannoutsos CT, Vinikoor M, van Lettow M, Nalugoda F, et al. (2019) Changes in rapid HIV treatment initiation after national "treat all" policy adoption in 6 sub-Saharan African countries: Regression discontinuity analysis. PLoS Med 16(6): e1002822. https://doi.org/10.1371/journal.pmed.1002822

Academic Editor: Marie-Louise Newell, University of Southampton, UNITED KINGDOM

Received: December 1, 2018

Accepted: May 10, 2019

Published: June 10, 2019

**Copyright:** © 2019 Tymejczyk et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Complete data for this study cannot be posted in a supplemental file or a public repository because of legal and ethical restrictions. The Principles of Collaboration under which the IeDEA multiregional collaboration was founded and the regulatory requirements of the different countries' IRBs require the submission and approval of a project concept sheet by the IeDEA Executive Committee and the principal investigators from participating regions. All datasets provided by IeDEA are deidentified RESEARCH ARTICLE

# Changes in rapid HIV treatment initiation after national "treat all" policy adoption in 6 sub-Saharan African countries: Regression discontinuity analysis

Olga Tymejczyk<sup>1,2\*</sup>, Ellen Brazier<sup>1,2</sup>, Constantin T. Yiannoutsos<sup>3</sup>, Michael Vinikoor<sup>4,5</sup>, Monique van Lettow<sup>6,7</sup>, Fred Nalugoda<sup>8</sup>, Mark Urassa<sup>9</sup>, Jean d'Amour Sinayobye<sup>10</sup>, Peter F. Rebeiro<sup>11</sup>, Kara Wools-Kaloustian<sup>12</sup>, Mary-Ann Davies<sup>13</sup>, Elizabeth Zaniewski<sup>14</sup>, Nanina Anderegg<sup>14</sup>, Grace Liu<sup>1,2</sup>, Nathan Ford<sup>15</sup>, Denis Nash<sup>1,2\*</sup>, on behalf of the leDEA consortium<sup>1</sup>

Institute for Implementation Science in Population Health, City University of New York, New York, New York, United States of America, 2 Graduate School of Public Health and Health Policy, City University of New York, New York, New York, United States of America, 3 Richard M. Fairbanks School of Public Health, Indiana University, Indianapolis, Indiana, United States of America, 4 Centre for Infectious Disease Research in Zambia, Lusaka, Zambia, 5 Department of Medicine, University of Alabama, Birmingham, Alabama, United States of America, 6 Dignitas International, Zomba, Malawi, 7 Dalla Lana School of Public Health, University of Toronto, Toronto, Ontario, Canada, 8 Rakai Health Sciences Program, Kalisizo and Entebbe, Uganda,
 Mwanza Intervention Trials Unit, National Institute for Medical Research, Mwanza, Tanzania, 10 Rwanda Military Hospital, Kigali, Rwanda, 11 Vanderbilt University School of Medicine, Nashville, Tennessee, United States of America, 12 Division of Infectious Diseases, Indiana University School of Medicine, Indianapolis, Indiana, United States of America, 13 Centre for Infectious Disease Epidemiology and Research, School of Public Health and Family Medicine, University of Cape Town, Cape Town, South Africa, 14 Institute of Social and Preventive Medicine, University of Bern, Bern, Switzerland, 15 Global Hepatitis Programme, HIV/AIDS Department, World Health Organization, Geneva, Switzerland

¶ Full membership of the leDEA consortium is provided in S1 Text.
\* olga.tymejczyk@sph.cuny.edu (OT); denis.nash@sph.cuny.edu (DN)

# Abstract

# Background

Most countries have formally adopted the World Health Organization's 2015 recommendation of universal HIV treatment ("treat all"). However, there are few rigorous assessments of the real-world impact of treat all policies on antiretroviral treatment (ART) uptake across different contexts.

# Methods and findings

We used longitudinal data for 814,603 patients enrolling in HIV care between 1 January 2004 and 10 July 2018 in 6 countries participating in the global International epidemiology Databases to Evaluate AIDS (IeDEA) consortium: Burundi (N = 11,176), Kenya (N = 179,941), Malawi (N = 84,558), Rwanda (N = 17,396), Uganda (N = 96,286), and Zambia (N = 425,246). Using a quasi-experimental regression discontinuity design, we assessed the change in the proportion initiating ART within 30 days of enrollment in HIV care (rapid ART initiation) after country-level adoption of the treat all policy. A modified Poisson model was

according to HIPAA Safe Harbor guidelines, with the exception of dates in some of the regions. IeDEA promotes the signing of a Data Use Agreement before HIV clinical data can be released. To request data, readers may contact IeDEA for consideration and instructions by filling out the online form available at www.iedea.org/home/whowe-are and completing the application at www. iedea.org/wp-content/uploads/2017/05/IeDEA\_ Multiregional\_Concept\_Application\_Form\_August\_ 2016.docx.

Funding: Research reported in this publication was supported by the National Institute of Allergy and Infectious Diseases [www.niaid.nih.gov] of the National Institutes of Health under Award Numbers U01AI096299 (OT. EB. JDS. GL. DN: leDEA Central Africa) and U01AI069924 (MD, EZ, NA, MV, MvL); IeDEA Southern Africa); the National Institute of Allergy and Infectious Diseases, Eunice Kennedy Shriver National Institute of Child Health and Human Development, the National Institute on Drug Abuse [www.drugabuse.gov], the National Cancer Institute, and the National Institute of Mental Health under Award Number U01AI069911 (CTY, KWK, MU, FN); leDEA East Africa); the Eunice Kennedy Shriver National Institute of Child Health and Human Development, the National Cancer Institute, the National Institute of Allergy and Infectious Diseases, National Institute of Mental Health, and the Office of the Director, the National Institutes of Health [www.nih.gov/ institutes-nih/nih-office-director], under Award Number U01AI069923 (Caribbean, Central and South America network for HIV epidemiology [CCASAnet]; PFR), and National Institute of Allergy and Infectious Diseases under Award Number K01-AI131895 (PFR). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

Abbreviations: aRR, adjusted risk ratio; ART, antiretroviral therapy; IeDEA, International epidemiology Databases to Evaluate AIDS; PLWH, people living with HIV; pp, percentage point; UNAIDS, Joint United Nations Programme on HIV/ AIDS; WHO, World Health Organization. used to identify factors associated with failure to initiate ART rapidly under treat all. In each of the 6 countries, over 60% of included patients were female, and median age at enrollment ranged from 32 to 36 years. In all countries studied, national adoption of treat all was associated with large increases in rapid ART initiation. Significant increases in rapid ART initiation immediately after treat all policy adoption were observed in Rwanda, from 44.4% to 78.9% of patients (34.5 percentage points [pp], 95% CI 27.2 to 41.7; p < 0.001), Kenya (25.7 pp, 95% CI 21.8 to 29.5; p < 0.001), Burundi (17.7 pp, 95% CI 6.5 to 28.9; p = 0.002), and Malawi (12.5 pp, 95% CI 7.5 to 17.5; p < 0.001), while no immediate increase was observed in Zambia (0.4 pp, 95% CI -2.9 to 3.8; p = 0.804) and Uganda (-4.2 pp, 95% CI -9.0 to 0.7; p = 0.090). The rate of rapid ART initiation accelerated sharply following treat all policy adoption in Malawi, Uganda, and Zambia; slowed in Kenya; and did not change in Rwanda and Burundi. In post hoc analyses restricted to patients enrolling under treat all, young adults (16-24 years) and men were at increased risk of not rapidly initiating ART (compared to older patients and women, respectively). However, rapid ART initiation following enrollment increased for all groups as more time elapsed since treat all policy adoption. Study limitations include incomplete data on potential ART eligibility criteria, such as clinical status, pregnancy, and enrollment CD4 count, which precluded the assessment of rapid ART initiation specifically among patients known to be eligible for ART before treat all.

#### Conclusions

Our analysis indicates that adoption of treat all policies had a strong effect on increasing rates of rapid ART initiation, and that these increases followed different trajectories across the 6 countries. Young adults and men still require additional attention to further improve rapid ART initiation.

## Author summary

# Why was this study done?

- Since late 2015, the World Health Organization (WHO) has recommended that all people living with HIV (PLWH) initiate ART, regardless of disease stage and CD4 cell count, and as of mid-2018, 84% of low- and middle-income countries had formally adopted WHO's recommendation to provide universal treatment to all PLWH.
- While modeling studies and trials have indicated that universal treatment, or "treat all," increases rapid ART initiation and viral load suppression among patients, there is little evidence on the impact of treat all policies on uptake of ART in real-world implementation settings.

## What did the researchers do and find?

• We assessed rapid ART initiation (i.e., ART initiation within 30 days of enrolling into HIV care) among 814,603 patients before and after national adoption of treat all policies in 6 countries in sub-Saharan Africa (Burundi, Kenya, Malawi, Rwanda, Uganda, and Zambia).

- To compare rapid ART initiation among those enrolling into care just before and just after national adoption of the treat all policy, we used a regression discontinuity design, using the date of HIV care enrollment as a continuous eligibility assignment variable. This design creates a quasi-experimental condition where the only systematic difference between patients enrolling just before and just after treat all policy adoption is the probability of treatment eligibility.
- There were large increases in rapid ART initiation after national adoption of treat all, with 81.6% of patients initiating ART within 30 days of enrollment after the policy adoption, and little difference in rapid ART initiation across patients with different immuno-deficiency status, as measured by CD4 counts.
- In Burundi, Kenya, Malawi, and Rwanda, there were large and significant increases in rapid ART initiation immediately following national adoption of treat all, ranging from 12.5- and 17.7-percentage-point increases in rapid ART initiation in Malawi and Burundi, respectively, to 25.7- and 34.5-percentage-point increases in Kenya and Rwanda.
- In Uganda and Zambia, there was no significant increase in rapid ART initiation immediately following national adoption of treat all. However, there were significant increases in the rate of rapid ART initiation in the months following treat all adoption, with each additional month following the policy adoption being associated with 2.2- and 2.6-percentage-point increases in the proportion of patients initiating ART in Uganda and Zambia, respectively.
- In the period following treat all adoption, men and young adults (aged 16–24 years) were less likely to rapidly initiate ART than women and older adults, respectively. However, for all groups, the risk of failing to rapidly initiate ART significantly decreased with elapsed time after treat all policy adoption.

#### What do these findings mean?

- Using a regression discontinuity design and real-world service delivery data, primarily from public-sector health facilities, our findings suggest that national adoption of treat all policies has led to appreciable increases in rapid ART initiation.
- As more follow-up data become available, future research should assess longer-term outcomes, such as retention in care, viral suppression, and treatment failure among patients initiating therapy under treat all.

## Introduction

In September 2015, after a series of clinical-stage- and CD4-count-based antiretroviral therapy (ART) eligibility expansions [1], the World Health Organization (WHO) recommended that all people living with HIV (PLWH) should start ART regardless of disease stage and at any CD4 cell count [2]. In 2017, WHO recommended treatment initiation within 7 days of confirming HIV diagnosis, including same-day ART initiation when feasible [3]. This recommendation followed from the results of several randomized trials and observational studies showing that rapid ART initiation was associated with reduced mortality and morbidity, and with a greater likelihood of achieving virological suppression and retention in care [4]. If initiated rapidly, universal HIV treatment (also known as "treat all" and "test and start") is thus an important strategy for reaching the Joint United Nations Programme on HIV/AIDS (UNAIDS) 90-90-90 targets.

Implementation of treat all is accelerating, and as of mid-2018, 84% of low- and middleincome countries had formally adopted the recommendation to provide universal treatment to PLWH [5]. Given the relatively recent rollout of treat all policies at the country level, however, there is little evidence to date on the policy's real-world effect on ART uptake. As the number of patients on ART grows under treat all policies, it is also critical to examine how the policy impacts patient groups that have historically lagged behind in HIV care engagement (i.e., men and young adults). While changes in ART initiation following WHO's 2009 and 2013 ART eligibility expansion recommendations have been evaluated [6–8], most estimates of the potential impact of treat all to date come from trials and modeling studies [9–13]. Although most of these research studies have reported favorable outcomes, they may not be generalizable to other contexts, including settings where health system and other resource constraints contribute to delays in policy implementation.

In this study, we used longitudinal patient data from 6 countries participating in the International epidemiology Databases to Evaluate AIDS (IeDEA) consortium to describe ART initiation rates under different country-level eligibility guidelines, to assess the impact of treat all policies on rapid ART initiation, and to identify factors associated with failure to start treatment rapidly under treat all.

#### Methods

#### Data sources and management

**Patient data.** IeDEA (https://www.iedea.org) captures demographic and clinical data on over 1.7 million patients receiving HIV care in 46 countries across 7 regional cohort collaborations [14]. The data represent a diverse cross-section of PLWH and HIV treatment programs, the majority of which (87%) are at public-sector health facilities, including both primary (42%) and secondary/tertiary-level sites (58%) [15]. In this analysis, we used medical records from patients enrolled in these programs between 1 January 2004 and 10 July 2018 in 6 countries in 3 regional cohorts with post-treat-all data available for analysis (Burundi and Rwanda in Central Africa, Kenya and Uganda in East Africa, and Malawi and Zambia in Southern Africa). Prior to merging and analysis, each region's data were standardized by regional data managers in accordance with IeDEA data exchange standards for variable definitions and data formatting.

**ART guidelines.** For each country in the analysis, we identified the dates of major ART eligibility expansions to  $CD4 \le 350$  cells/µl,  $CD4 \le 500$  cells/µl, and treat all. We have previously described our systematic search for current and historical ART eligibility guidelines based on publicly available policy documents, published literature, and inputs from in-country experts [6]. If the exact date of ART eligibility expansion was unknown, it was assumed to have occurred on the first day of the month in which the policy was adopted. In Malawi, the date of national treat all adoption published in the country's guidelines was adjusted to 2 months later to reflect the delays in policy rollout documented by the Ministry of Health [16].

The study utilized de-identified data approved for use by local ethical committees in each of the IeDEA regions included in the analysis.

#### Inclusion criteria

Patients had to be at least 16 years old at enrollment and have at least 30 days of possible follow-up between enrollment and database closure. Patients were excluded if they transferred to an IeDEA site from another clinic or were known not to be ART naïve at enrollment (with ART defined as any regimen of at least 3 antiretroviral drugs, excluding treatment taken solely for prevention of mother-to-child transmission). We excluded sites with no patient data available for the period between care enrollment and ART initiation (i.e., pre-ART data, such as documented visits and laboratory tests prior to ART initiation), as well as sites that only reported data for ART initiators (defined as sites where fewer than 2% of patients over the study period never initiated ART).

#### Outcome and exposure

The outcome of interest was rapid ART initiation, defined as initiation of treatment within 30 days of enrollment in HIV care (distinct from the 2017 WHO definition of rapid ART initiation as occurring within 7 days of HIV diagnosis) [3]. The exposure was time of enrollment in HIV care with respect to the calendar date of country-level ART eligibility expansion to treat all.

#### Other definitions

To represent pre-treatment HIV disease severity, CD4 count at enrollment was defined as the CD4 measurement closest to the enrollment date within a  $\pm$ 90-day window, but no later than 1 week after ART initiation.

#### Study design

**Descriptive analyses.** The proportion of patients initiating ART rapidly in each of the 4 ART eligibility periods (at  $CD4 \le 200$  or  $CD4 \le 250$  cells/µl [period 1]; at  $CD4 \le 350$  cells/µl [period 2]; at  $CD4 \le 500$  cells/µl [period 3]; and treat all [period 4]) was calculated and stratified by country, sex, and age group (dichotomized as 16–24 or  $\ge 25$  years to approximate the 15–24-year age category commonly used by UNAIDS) [17].

Effect of ART eligibility expansion to treat all on rapid ART initiation. A regression discontinuity design was applied to assess the effect of enrollment in HIV care under treat all on the proportion of patients initiating ART rapidly. This approach takes advantage of the local randomness around a cutoff-based, continuous eligibility assignment variable (in this case, calendar date of HIV care enrollment relative to the date of country-level adoption of treat all). This local randomness creates a quasi-experimental condition in which there are no systematic differences between patients enrolling in care on either side of the cutoff threshold (date of country-level treat all adoption), but patients on one side (those enrolling after treat all adoption) have a higher probability of initiating treatment than on the other, permitting a causal interpretation of observed effects [18–20]. Covariate balance tests for patients enrolling immediately before and after treat all adoption and plots of the assignment variable were completed to assess the possibility of systematic differences between patients enrolling in HIV care on either side of the threshold, as well as nonrandom enrollment before or after the treat all adoption date.

As information about each patient's ART eligibility at enrollment (e.g., HIV stage, comorbidity, pregnancy, and/or key population status) was incomplete, the study is an intent-totreat analysis using a "sharp" regression discontinuity design [18,19]. For each of the 6 countries in the analysis, we examined the unadjusted association between calendar date of enrollment in HIV care and rapid ART initiation. A discontinuity at the threshold of the date of national treat all policy adoption allowed for different slopes before and after the cutoff date. To estimate predicted outcomes and risk differences at the treat all threshold date, the following local linear regression models [21] were used:

$$E[Y_i|Z_i] = \beta_0 + \beta_1 * Z_i + \beta_2 * 1[Z_i \ge 0] + \beta_3 * Z_i * 1[Z_i \ge 0]$$

where  $Y_i$  is the patient-level outcome (rapid ART initiation),  $Z_i$  is the number of days between a patient's enrollment date and the national treat all policy introduction date (negative if patient enrolled before the policy was introduced), and  $1[Z_i \ge 0]$  indicates whether a patient enrolled after the policy was introduced or not.

Data-driven Imbens–Kalyanaraman bandwidths were applied to minimize the mean squared error of the difference in predicted values at the threshold date of treat all introduction [22]. All observations within the bandwidth window were weighted equally (rectangular kernel). Sensitivity analyses were completed using 3 other bandwidth sizes (100, 200, and 300 days). In order to exclude possible threshold effects of prior eligibility expansions, only data for patients enrolling at least 90 days after the preceding ART eligibility expansion were included in bandwidth calculations and subsequent treat all regression discontinuity analysis. Prior to bandwidth calculations, patients enrolling in the 30 days immediately before treat all adoption were also excluded to ensure that there was no overlap in the outcome estimation windows (i.e., 30 days following enrollment) of those enrolling before and after the adoption of treat all introduction), 30 days were added to remaining enrollment dates preceding policy introduction. In effect, a patient who enrolled 31 days prior to policy introduction had  $Z_i = -1$ .

Pooled estimates of the risk difference at the treat all threshold were obtained from metaanalysis using a DerSimonian and Laird random-effects model, which computed a weighted average of all 6 countries' effect estimates based on their standard errors [23].

Trends in rapid ART initiation before and after treat all adoption. To characterize trends in rapid ART initiation, slopes from linear regression models for the period before (starting 90 days after ART eligibility expansion to  $CD4 \leq 500$  cells/µl) and after the date of treat all adoption were compared and expressed as the percentage point (pp) change in rapid ART initiation per month.

**Correlates of failure to initiate ART rapidly under treat all.** To identify factors associated with failure to initiate ART rapidly among patients enrolling in HIV care under treat all, a multivariable, modified Poisson model with robust error variances was used [24]. A single model was fitted for all 6 countries, with age group, sex, and time between treat all adoption and enrollment (categorized into 0 to <3 months, 3 to <6 months, 6 to <12 months, and  $\geq$ 12 months) as covariates. To represent other, unobserved confounders and health system features, the model also included a variable for country. Enrollment CD4 count was not included in the model, as it was only available for 37% of the patients.

Descriptive analyses were completed in SAS 9.3 and regression discontinuity analyses in Stata/IC version 14.2 [25].

The analysis plan was outlined in a May 2018 proposal to the IeDEA Executive Committee and is documented in <u>S2 Text</u>. The analysis of correlates of failure to initiate ART rapidly under treat all was not pre-specified.

#### Results

#### Sample characteristics

The 6 sub-Saharan African countries in the analysis adopted treat all policies between July and December 2016. A total of 814,603 patients met the study's inclusion criteria, ranging from 11,176 in Burundi to 425,246 in Zambia. Overall, median age at enrollment across countries was 33 years. Over 60% of HIV care enrollees in all 6 countries were women, though the proportion decreased appreciably over time in Kenya and Rwanda (from 67.6% to 59.7% and from 64.8% to 56.7%, respectively, from period 1 to period 4) (Table 1).

Distribution of baseline characteristics among newly enrolling patients was similar just before and just after treat all adoption (excluding the 30 days immediately preceding the policy change) (S1 Table). No major discontinuity in the number of new enrollments around the date of treat all adoption was observed, consistent with our efforts to exclude sites with no pre-ART patient data (i.e., treat all did not result in an artificial increase in newly enrolling patients) (S1 Fig).

# Rapid ART initiation under treat all versus in previous ART eligibility periods (descriptive analysis)

Across the 6 countries, 81.6% of patients enrolling under treat all initiated treatment rapidly (within 30 days); 59.2% initiated on the day of enrollment in HIV care, and 67.1% within 7 days. Rapid ART initiation under treat all was highest in Malawi (88.9%) and Rwanda (86.9%), and lowest in Burundi (77.9%) (Table 2).

Overall, rapid ART initiation increased by 25.9 pp under treat all, compared with the  $CD4 \leq 500$  cells/µl eligibility period (81.6% versus 55.7%). Larger increases in rapid ART initiation under treat all were observed among patients  $\geq$ 25 years old than among those aged 16–24 years (26.5 versus 22.9 pp), whereas following the expansion to ART eligibility at  $CD4 \leq 500$  cells/µl, larger increases in rapid ART initiation were observed among those aged 16–24 years (23.9 versus 17.2 pp). No appreciable differences by sex were observed. The greatest increase in the proportion of patients initiating ART rapidly following expansion to treat all was observed in Rwanda (47.2 pp, from 39.7% to 86.9%) and Kenya (33.8 pp, from 49.3% to 83.1%) (S2 Fig).

Increases in rapid ART initiation were observed at every enrollment CD4 count level after each ART eligibility expansion. For earlier ART eligibility expansions, there were large differences in the proportion initiating ART among those with CD4 counts above the treatment eligibility threshold and those with CD4 counts below the threshold. For example, after eligibility expansion to CD4  $\leq$  500 cells/µl, there was a 17.6-pp difference in rapid ART initiation between patients enrolling with CD4 counts of 451 to 500 cells/µl and those with CD4 counts of 501 to 550 cells/µl (58.9% versus 41.3%; medium blue points in Fig 1). In contrast, under treat all, there was little difference in rapid ART initiation across CD4 count levels, as indicated by the lack of any inflection point (dark blue points in Fig 1).

# Effect of treat all adoption on rapid ART initiation (regression discontinuity analysis)

In 4 of the 6 countries, there was a statistically significant increase in rapid ART initiation immediately following national adoption of treat all. The effect was greatest in Rwanda, with a 34.5-pp increase (95% CI 27.2 to 41.7): 78.9% of patients enrolling immediately after treat all adoption initiated ART rapidly, compared with 44.4% of patients enrolling immediately before the policy change (a 77.7% relative increase). Increases were more moderate in Kenya

#### Table 1. Baseline characteristics of patients enrolling in all 4 ART eligibility periods between 2004 and 2018 (N = 814,603).

Characteristic	Overall	Period 1 (CD4 ≤ 200/250 cells/μl)	Period 2 (CD4 ≤ 350 cells/µl)	Period 3 (CD4 $\leq$ 500 cells/µl)	Period 4 (treat all)
Total N	814,603	389,416	216,832	169,822	38,533
Burundi					
Date of guideline expansion			October 2010	August 2014	September 2016
N	11,176	5,883	3,031	1,233	1,029
Median age (IQR) (years)	36 (28-44)	36 (29-45)	35 (28-44)	36 (27-45)	36 (27-45)
Percent female	63.8	65.7	61.6	63.3	59.7
Percent with enrollment CD4 count	19.1	13.3	33.6	16.0	13.4
Median enrollment CD4 count (IQR) (cells/µl)	302 (169-495)	257 (151-417)	323 (174–521)	347 (178–531)	432 (250-653)
Kenya					
Date of guideline expansion			November 2011	June 2014	July 2016
N	179,941	114,230	32,830	22,645	10,236
Median age (IQR) (years)	34 (27-41)	34 (28-42)	32 (26-40)	32 (26-41)	34 (28-43)
Percent female	66.5	67.6	66.0	64.6	59.7
Percent with enrollment CD4 count	73.5	81.4	69.7	57.3	32.6
Median enrollment CD4 count (IQR) (cells/µl)	242 (98-435)	228 (92-417)	267 (107-461)	302 (135-498)	272 (114–479)
Malawi					
Date of guideline expansion			July 2011	April 2014	July* 2016
N	84,558	37,215	24,803	18,881	3,659
Median age (IQR) (years)	32 (27-39)	33 (27-40)	32 (26–38)	32 (26–39)	32 (26-39)
Percent female	63.5	61.9	66.3	63.3	61.6
Percent with enrollment CD4 count	34.7	42.4	30.9	30.8	3.0
Median enrollment CD4 count (IQR) (cells/µl)	221 (115-352)	187 (93–295)	261 (146-391)	283 (156-431)	244 (154–411)
Rwanda					
Date of guideline expansion			August 2007	August 2011	July 2016
N	17,396	5,477	5,581	5,048	1,290
Median age (IQR) (years)	32 (26–39)	34 (28-40)	31 (26–38)	31 (25–38)	32 (26-39)
Percent female	61.5	64.8	59.3	61.7	56.7
Percent with enrollment CD4 count	75.3	67.2	79.8	79.9	72.2
Median enrollment CD4 count (IQR) (cells/µl)	350 (189-548)	278 (140-489)	364.5 (213-558)	384.5 (223-574)	391 (215-582)
Uganda					
Date of guideline expansion			June 2011	December 2013	November 2016
Ν	96,286	53,028	20,956	17,909	4,393
Median age (IQR) (years)	32 (26-39)	33 (27-40)	30 (25–38)	30 (24–37)	30 (24–37)
Percent female	62.8	64.2	61.7	60.5	61.3
Percent with enrollment CD4 count	72.5	69.7	79.3	78.7	49.3
Median enrollment CD4 count (IQR) (cells/µl)	278 (122-477)	236 (95–438)	315 (155–502)	337 (170–516)	334 (150–529)
Zambia					
Date of guideline expansion			June 2010	December 2013	December 2016
N	425,246	173,583	129,631	104,106	17,926
Median age (IQR) (years)	33 (28-40)	33 (28–40)	33 (27-40)	33 (27-40)	32 (26–39)
Percent female	62.0	62.9	61.6	61.3	60.8
Percent with enrollment CD4 count	67.2	79.8	66.7	52.4	34.0
Median enrollment CD4 count (IQR) (cells/µl)	226 (111-385)	195 (94-342)	253 (129-410)	270 (135-436)	289 (149-469)

\*Adjusted based on documented policy rollout delays (see Methods).

https://doi.org/10.1371/journal.pmed.1002822.t001

Sample	Number of enrollees under treat all	Period of enrollment	Proportion of patients initiating ART			
			On the day of enrollment	Within 7 days of enrollment	Within 30 days of enrollment	
Overall	38,533	May 2016–Jul 2018	59.2%	67.1%	81.6%	
By country						
Burundi	1,029	Sep 2016–Jul 2018	32.7%	55.4%	77.9%	
Kenya	10,236	Jul 2016–Jan 2018	63.6%	67.1%	83.1%	
Malawi	3,659	Jul 2016-Nov 2016	69.5%	82.5%	88.9%	
Rwanda	1,290	Jul 2016–Dec 2017	25.7%	56.7%	86.9%	
Uganda	4,393	Nov 2016–Jan 2018	66.3%	71.7%	81.2%	
Zambia	17,926	Dec 2016-Aug 2017	56.7%	64.2%	79.1%	

Table 2. Proportion of patients enrolling in HIV care under treat all and initiating ART on the day of enrollment, within 7 days, and within 30 days ("rapid ART initiation").

https://doi.org/10.1371/journal.pmed.1002822.t002

(25.7 pp, 95% CI 21.8 to 29.5), Burundi (17.7 pp, 95% CI 6.5 to 28.9), and Malawi (12.5 pp, 95% CI 7.5 to 17.5). No statistically significant discontinuity effect was observed at the treat all threshold in Uganda (-4.2 pp, 95% CI -9.0 to 0.7) or Zambia (0.4 pp, 95% CI -2.9 to 3.8). The



Fig 1. Rapid ART initiation (within 30 days of enrollment) across ART eligibility periods and enrollment CD4 counts (cells/µl).

https://doi.org/10.1371/journal.pmed.1002822.g001

Measure	Pooled	Rwanda	Kenya	Burundi	Malawi	Zambia	Uganda
Risk difference at the treat all adoption threshold*	14.2	34.5	25.7	17.7	12.5	0.4	-4.2
95% CI	(2.2, 26.2)	(27.2, 41.7)	(21.8, 29.5)	(6.5, 28.9)	(7.5, 17.5)	(-2.9, 3.8)	(-9.0, 0.7)
<i>p</i> -Value	0.020	< 0.001	< 0.001	0.002	< 0.001	0.804	0.090
Imbens–Kalyanaraman bandwidth, days		405	158	376	97	79	205
N within bandwidth		2,189	8,052	1,163	5,020	12,762	5,371
Predicted outcomes at the treat all threshold*							
Enrollment just before treat all adoption		44.4%	55.0%	55.7%	68.6%	62.4%	70.6%
Enrollment just after treat all adoption		78.9%	80.7%	73.4%	81.1%	62.8%	66.4%
Relative change after treat all adoption		77.7%	46.7%	31.8%	18.2%	0.6%	-5.9%
Slopes before and after treat all adoption**							
Percentage point change in rapid ART initiation per month before treat all adoption		0.4	0.6	0.6	0.1	0.2	0.4
Percentage point change in rapid ART initiation per month after treat all adoption		0.8	0.3	0.7	2.8	2.6	2.2
<i>p</i> -Value for difference of slopes		0.105	0.006	0.619	< 0.001	< 0.001	< 0.001

Table 3. Effect of enrollment under treat all on rapid ART initiation (within 30 days of enrollment), by country and pooled, and slopes before and after treat all adoption.

\*Risk difference and predicted outcomes at the treat all threshold are from regression discontinuity analysis. Effects are calculated at the modified guideline expansion threshold of 31 days before versus 1 day after treat all adoption.

\*\*Slope comparison is from separate linear regression models comparing the period before treat all adoption (starting 90 days after ART eligibility expansion to  $CD4 \leq 500$  cells/µl) and after adoption.

https://doi.org/10.1371/journal.pmed.1002822.t003

pooled estimate of effect at the treat all threshold was 14.2 pp (95% CI 2.2 to 26.2) (Table 3; Fig 2).

Results of sensitivity analyses using other bandwidths were consistent with the findings based on the data-driven Imbens–Kalyanaraman bandwidth (S2 Table).

## Trends in rapid ART initiation before and after treat all adoption

After treat all adoption, the rate of change in rapid ART initiation increased most in Malawi (from 0.1 pp per month before treat all adoption to 2.8 pp per month afterwards), Zambia (from 0.2 pp to 2.6 pp), and Uganda (from 0.4 pp to 2.2 pp; all p < 0.001). A decrease in the rate was observed in Kenya (from 0.6 pp to 0.3 pp per month; p = 0.006).

## Correlates of failure to initiate ART rapidly under treat all

In post hoc analyses employing multivariable models restricted to patients enrolling under treat all, persons ages 16–24 years had an 18% greater risk of not starting ART rapidly, compared with those  $\geq$ 25 years old (adjusted risk ratio [aRR] = 1.18, 95% CI 1.12 to 1.24), and men had a 12% higher risk than women (aRR = 1.12, 95% CI 1.07 to 1.17). There was an inverse relationship between time elapsed since treat all adoption and failure to initiate ART rapidly: At 12 months after treat all adoption, new enrollees had 47% lower risk of not starting ART rapidly compared to those who enrolled in the first 3 months after the policy was adopted (aRR = 0.53, 95% CI 0.48 to 0.58) (Table 4).

## Discussion

We found that adoption of treat all policies at the national level in 6 African countries was followed by appreciable increases in the proportion of patients initiating ART rapidly (within 30





\*Adjusted based on documented policy rollout delays (see Methods).

https://doi.org/10.1371/journal.pmed.1002822.g002

days of enrollment) in diverse, largely public-sector service delivery settings. Regression discontinuity analysis found marked increases in rapid ART initiation in most of the countries immediately after the national adoption of treat all. Our study provides evidence, outside of a controlled research environment, of a causal effect of national adoption of the treat all policy

#### Table 4. Correlates of failure to initiate ART rapidly (within 30 days of enrollment) under treat all (N = 38,533).

Characteristic	N	RR (95% CI)	p-Value	aRR (95% CI)	<i>p</i> -Value
Age					
16-24 years	7,259	1.15 (1.10, 1.21)	< 0.001	1.18 (1.12–1.24)	<0.001
$\geq$ 25 years	31,274	Ref		Ref	
Sex					
Male	15,220	1.08 (1.04, 1.13)	< 0.001	1.12 (1.07–1.17)	<0.001
Female	23,313	Ref		Ref	
Time between treat all ado	ption and enrollmen	t			
0 to <3 months	14,075	Ref	< 0.001	Ref	<0.001
3 to <6 months	12,035	0.74 (0.70, 0.77)		0.71 (0.68-0.75)	
6 to <12 months	8,737	0.67 (0.63, 0.71)		0.63 (0.59-0.66)	
$\geq$ 12 months	3,686	0.57 (0.53, 0.63)		0.53 (0.48-0.58)	

#### Model also adjusted for country.

aRR, adjusted risk ratio; RR, risk ratio.

https://doi.org/10.1371/journal.pmed.1002822.t004

on sustained improvements in rapid ART initiation following HIV care enrollment, with no apparent negative effects on ART initiation for those eligible under prior treatment guidelines.

In half of the countries in the analysis, immediate relative increases in rapid ART initiation at the regression discontinuity threshold (ranging from 31.8% to 77.7% in Burundi, Kenya, and Rwanda) were larger than the 26.7% increase in the number of ART initiators predicted for a 1-year period after treat all adoption by a recent empirical modeling study for South Africa [26], and similar to those observed in a recent trial in Eswatini [12] and a recent analysis of ART uptake in Rwanda following national adoption of treat all [27].

Improvements were sustained or further amplified following national adoption of treat all. In Malawi, Uganda, and Zambia, the rate of change in rapid ART initiation increased sharply in the months following the date of treat all adoption, suggesting that implementation of expanded treatment eligibility guidelines may have been phased in or delayed. This was particularly true in Uganda and Zambia, where no effect was observed at the threshold. Lack of an immediate effect may also suggest that there were few newly eligible patients at the time when treat all was adopted, or it may reflect regression to the mean, given already-high levels of rapid ART initiation in the months leading up to eligibility expansion.

Previous studies have reported lower rates of ART initiation among men [9,28] and young adults [6,29], compared to women and older adults. Similarly, while rapid ART initiation rates improved for both groups following the adoption of treat all, our analysis found that men and young adults remained at greater risk of not initiating ART rapidly. While we found that the risk of failing to rapidly initiate ART decreased markedly with time following national adoption of treat all, as more follow-up data become available, future research should assess whether age and sex disparities in ART initiation diminish over time.

Overall, there were no appreciable differences in rates of rapid ART initiation across different enrollment CD4 counts. If observed, such differences would have suggested that the sickest patients were being "crowded out" by patients with less advanced disease under treat all. While these findings are consistent with research assessing the impacts of prior ART eligibility expansions [6,7,30], there may be variations at the country or site level that we were unable to assess in stratified analyses, largely because substantially fewer patients under treat all have enrollment CD4 counts measured before ART initiation.

A major strength of our study is the use of a regression discontinuity design and service delivery data that reflect diverse real-world implementation settings in sub-Saharan Africa, which provide support for a causal interpretation of the association between expanded ART eligibility under treat all and increases in rapid ART uptake [20]. The use of data-driven Imbens–Kalyanaraman optimal bandwidths and sensitivity analyses with 3 other constant bandwidths enabled us to generate robust effect estimates with minimal risk of researcher bias [22].

However, a limitation of our study is incomplete data on potential ART eligibility criteria, such as clinical status (including TB coinfection) and pregnancy, as well as enrollment CD4 count. Such data would have better elucidated the actual ART eligibility status of patients enrolling before treat all. These gaps precluded the implementation of a "fuzzy" regression discontinuity design, which would better reflect the probabilistic distribution of ART eligibility in the pre-treat-all sample [18]. Similarly, we were unable to assess rapid ART initiation specifically among patients previously eligible for ART, which could have provided insights into barriers to rapid ART initiation in the pre-treat-all era (e.g., capacity constraints and/or delays in eligibility ascertainment versus adherence to policy at the time). In addition, the limited availability of data on patient characteristics other than age, sex, and enrollment CD4 count restricted our ability to assess whether patients on each side of the regression discontinuity threshold were similar with respect to other pre-treatment covariates.

Another limitation is that the exact date of treat all policy adoption at the site level is unknown for many IeDEA sites, and delays in site-level implementation of expanded treatment eligibility guidelines likely varied within and across countries. Such potential non-differential exposure misclassification would be expected to attenuate observed effects and may have contributed to the observed results in Uganda and Zambia (i.e., no increase in rapid ART initiation at the regression discontinuity threshold, but increases in ART initiation rates with increasing time after national adoption of treat all). At the same time, if facilities contributing data to an international collaboration such as IeDEA have a greater capacity to rapidly implement policy changes, the threshold effects we found may overestimate the effects of the adoption of treat all in the respective countries.

These findings are nonetheless important, given previous research showing that loss to care between diagnosis and ART initiation is the most common "breakpoint" in the HIV care continuum [31], as well as recent evidence of improved retention in care among those who are immediately eligible for ART [4,11,19]. As more data on HIV care under treat all become available, research should assess longer-term patient outcomes, including rate of and time to viral load suppression among patients who immediately initiate ART. Recent findings from a trial in Eswatini have reported large improvements in viral load suppression under treat all [11]; however, additional analyses utilizing real-world service delivery data from diverse country contexts and quasi-experimental designs will be important for deriving generalizable effect estimates of the individual- and population-level benefits of treat all policies. Country-specific analyses incorporating multiple change points for treatment eligibility expansions could provide additional insight into the relative impacts of distinct policies over time in diverse contexts. Further research into health system constraints, demand-side barriers, and underserved populations is also critical for understanding the drivers of between-country differences in the effect of treat all policies on rapid ART initiation. Equally important is implementation research to identify effective strategies for increasing the uptake of HIV testing and linkage to care [26,32], as well as for optimizing patient outcomes along the HIV care cascade under treat all [33]. Finally, as more follow-up data under treat all become available, questions around retention and viral suppression, as well as treatment failure and subsequent regimen switches, will be important to examine.

In conclusion, our study demonstrates a strong and sustained effect of national-level adoption of treat all policies on rapid ART initiation in diverse service delivery settings across 6 sub-Saharan African countries. This provides further evidence of treat all being a key strategy towards the achievement of UNAIDS 90-90-90 targets.

## **Supporting information**

**S1 STROBE checklist.** (DOCX)

**S1** Fig. Distribution of HIV care enrollment in the year before and after treat all adoption. (TIF)

**S2 Fig. Rapid ART initiation (within 30 days of enrollment) across enrollment/patient characteristics.** (A) ART eligibility period, (B) age group, (C) sex, and (D) country. (TIF)

**S1** Table. Covariate balance at the treat all adoption threshold. (DOCX)

S2 Table. Sensitivity analyses with other bandwidth sizes. (DOCX)
S1 Text. Full membership of the IeDEA consortium. (DOCX)
S2 Text. Analysis concept sheet. (DOCX)

## Acknowledgments

The authors thank patients, providers, and administrative staff at participating IeDEA facilities. We also are grateful for information on ART eligibility expansions contributed by Marion Achieng, Dr. Anthony Harries, Dr. Sabin Nsanzimana, Dr. Erik Schouten, and Dr. Christella Twizere. Additionally, we thank Dr. Annette H. Sohn for her feedback on the manuscript. Full membership of the IeDEA consortium is provided in <u>S1 Text</u>.

## **Author Contributions**

Conceptualization: Olga Tymejczyk, Ellen Brazier, Nathan Ford, Denis Nash.

Data curation: Olga Tymejczyk, Constantin T. Yiannoutsos.

Formal analysis: Olga Tymejczyk.

Funding acquisition: Denis Nash.

**Investigation:** Olga Tymejczyk, Ellen Brazier, Michael Vinikoor, Monique van Lettow, Fred Nalugoda, Mark Urassa, Jean d'Amour Sinayobye, Peter F. Rebeiro, Kara Wools-Kaloustian, Mary-Ann Davies, Elizabeth Zaniewski, Nanina Anderegg, Grace Liu, Nathan Ford, Denis Nash.

Methodology: Olga Tymejczyk, Constantin T. Yiannoutsos, Peter F. Rebeiro, Denis Nash.

Resources: Denis Nash.

Supervision: Denis Nash.

Visualization: Olga Tymejczyk.

Writing - original draft: Olga Tymejczyk, Ellen Brazier.

Writing – review & editing: Olga Tymejczyk, Ellen Brazier, Constantin T. Yiannoutsos, Michael Vinikoor, Monique van Lettow, Fred Nalugoda, Mark Urassa, Jean d'Amour Sinayobye, Peter F. Rebeiro, Kara Wools-Kaloustian, Mary-Ann Davies, Elizabeth Zaniewski, Nanina Anderegg, Grace Liu, Nathan Ford, Denis Nash.

#### References

- 1. Eholié SP, Badje A, Kouame GM, N'takpe J-B, Moh R, Danel C, et al. Antiretroviral treatment regardless of CD4 count: the universal answer to a contextual question. AIDS Res Ther. 2016; 13(1):27.
- 2. World Health Organization. Guideline on when to start antiretroviral therapy and on pre-exposure prophylaxis for HIV. Geneva: World Health Organization; 2015.
- 3. World Health Organization. Guidelines for managing advanced HIV disease and rapid initiation of antiretroviral therapy. Geneva: World Health Organization; 2017.
- Ford N, Migone C, Calmy A, Kerschberger B, Kanters S, Nsanzimana S, et al. Benefits and risks of rapid initiation of antiretroviral therapy. AIDS. 2018; 32(1):17–23. https://doi.org/10.1097/QAD. 00000000001671 PMID: 29112073

- World Health Organization. WHO HIV policy adoption and implementation status in countries: fact sheet. Geneva: World Health Organization; 2018.
- Tymejczyk O, Brazier E, Yiannoutsos C, Wools-Kaloustian K, Althoff K, Crabtree-Ramirez B, et al. HIV treatment eligibility expansion and timely antiretroviral treatment initiation following enrollment in HIV care: a metaregression analysis of programmatic data from 22 countries. PLoS Med. 2018; 15(3): e1002534. https://doi.org/10.1371/journal.pmed.1002534 PMID: 29570723
- Kluberg SA, Fox MP, LaValley M, Pillay D, Barnighausen T, Bor J. Do ART eligibility expansions crowd out the sickest? Evidence from rural South Africa. Trop Med Int Health. 2018; 23(9):968–79. https://doi. org/10.1111/tmi.13122 PMID: 29947442
- Mody A, Sikazwe I, Czaicki NL, Wa Mwanza M, Savory T, Sikombe K, et al. Estimating the real-world effects of expanding antiretroviral treatment eligibility: evidence from a regression discontinuity analysis in Zambia. PLoS Med. 2018; 15(6):e1002574. https://doi.org/10.1371/journal.pmed.1002574 PMID: 29870531
- Petersen M, Balzer L, Kwarsiima D, Sang N, Chamie G, Ayieko J, et al. Association of implementation of a universal testing and treatment intervention with HIV diagnosis, receipt of antiretroviral therapy, and viral suppression in East Africa. JAMA. 2017; 317(21):2196–206. https://doi.org/10.1001/jama.2017. 5705 PMID: 28586888
- Kimmel A, Bono RS, Keiser O, Sinayobye JD, Estill J, Mujwara D, et al. Mathematical modeling to inform 'treat all' implementation in sub-Saharan Africa: a scoping review. J Virus Erad. 2018; 4(Suppl 2):47–54. PMID: 30515314
- 11. Khan S, Spiegelman D, Walsh F, Mazibuko S, Pasi M, Chai B, et al. Universal test and treat (UTT) versus standard of care for access to antiretroviral therapy in HIV clients: the MaxART stepped-wedge randomized controlled health systems trial in Swaziland. Abstract 13370. 22nd International AIDS Conference; 2018 Jul 23–27; Amsterdam, Netherlands.
- Khan S, Walsh F, Spiegelman D, Mazibuko S, Pasi M, Lejeune C, et al. Time to antiretroviral therapy initiation under universal test and treat strategy in Swaziland: a stepped-wedge implementation trial. Abstract 11492. 22nd International AIDS Conference; 2018 Jul 23–27; Amsterdam, Netherlands.
- Iwuji CC, Orne-Gliemann J, Larmarange J, Balestre E, Thiebaut R, Tanser F, et al. Universal test and treat and the HIV epidemic in rural South Africa: a phase 4, open-label, community cluster randomised trial. Lancet HIV. 2018; 5(3):e116–25. <u>https://doi.org/10.1016/S2352-3018(17)30205-9</u> PMID: 29199100
- International epidemiology Databases to Evaluate AIDS. Home. International epidemiology Databases to Evaluate AIDS; 2019 [cited 2019 May 16]. Available from: <a href="https://www.iedea.org/">https://www.iedea.org/</a>.
- Fritz CQ, Blevins M, Lindegren ML, Wools-Kaloutsian K, Musick BS, Cornell M, et al. Comprehensiveness of HIV care provided at global HIV treatment sites in the IeDEA consortium: 2009 and 2014. J Int AIDS Soc. 2017; 20(1):20933. https://doi.org/10.7448/IAS.20.1.20933 PMID: 28364561
- 16. Government of Malawi Ministry of Health. Integrated HIV program report July–September 2016. Lilongwe: Government of Malawi Ministry of Health; 2016.
- 17. Joint United Nations Programme on HIV/AIDS. Global AIDS update 2016. Geneva: Joint United Nations Programme on HIV/AIDS; 2016.
- Moscoe E, Bor J, Barnighausen T. Regression discontinuity designs are underutilized in medicine, epidemiology, and public health: a review of current and best practice. Clin Epidemiol. 2015; 68(2):122–33.
- Bor J, Fox MP, Rosen S, Venkataramani A, Tanser F, Pillay D, et al. Treatment eligibility and retention in clinical HIV care: a regression discontinuity study in South Africa. PLoS Med. 2017; 14(11): e1002463. https://doi.org/10.1371/journal.pmed.1002463 PMID: 29182641
- Bor J, Moscoe E, Mutevedzi P, Newell ML, Barnighausen T. Regression discontinuity designs in epidemiology: causal inference without randomized trials. Epidemiology. 2014; 25(5):729–37. https://doi.org/ 10.1097/EDE.00000000000138 PMID: 25061922
- Hahn J, Todd P, Van der Klaauw W. Identification and estimation of treatment effects with a regressiondiscontinuity design. Econometrica. 2001; 69(1):201–9.
- 22. Imbens G, Kalyanaraman K. Optimal bandwidth choice for the regression discontinuity estimator. Rev Econ Stud. 2012; 79(3):933–59.
- 23. DerSimonian R, Laird N. Meta-analysis in clinical trials. Control Clin Trials. 1986; 7(3):177–88. PMID: 3802833
- Zou G. A modified poisson regression approach to prospective studies with binary data. Am J Epidemiol. 2004; 159(7):702–6. https://doi.org/10.1093/aje/kwh090 PMID: 15033648
- 25. Calonico S, Cattaneo MD, Titiunik R. Robust data-driven inference in the regression-discontinuity design. Stata J. 2014; 14(4):909–46.

- 26. Bor J, Ahmed S, Fox MP, Rosen S, Meyer-Rath G, Katz IT, et al. Effect of eliminating CD4-count thresholds on HIV treatment initiation in South Africa: an empirical modeling study. PLoS ONE. 2017; 12(6): e0178249. https://doi.org/10.1371/journal.pone.0178249 PMID: 28617805
- Ross J, Sinayobye JD, Yotebieng M, Hoover DR, Shi Q, Ribakare M, et al. Early outcomes after implementation of treat all in Rwanda: an interrupted time series study. J Int AIDS Soc. 2019; 22(4):e25279. https://doi.org/10.1002/jia2.25279 PMID: 30993854
- Muula AS, Ngulube TJ, Siziya S, Makupe CM, Umar E, Prozesky HW, et al. Gender distribution of adult patients on highly active antiretroviral therapy (HAART) in Southern Africa: a systematic review. BMC Public Health. 2007; 7:63. https://doi.org/10.1186/1471-2458-7-63 PMID: 17459154
- Wong VJ, Murray KR, Phelps BR, Vermund SH, McCarraher DR. Adolescents, young people, and the 90-90-90 goals: a call to improve HIV testing and linkage to treatment. AIDS. 2017; 31(Suppl 3):S191– 4.
- Plazy M, Dabis F, Naidu K, Orne-Gliemann J, Barnighausen T, Dray-Spira R. Change of treatment guidelines and evolution of ART initiation in rural South Africa: data of a large HIV care and treatment programme. BMC Infect Dis. 2015; 15:452. https://doi.org/10.1186/s12879-015-1207-2 PMID: 26497054
- Levi J, Raymond A, Pozniak A, Vernazza P, Kohler P, Hill A. Can the UNAIDS 90-90-90 target be achieved? A systematic analysis of national HIV treatment cascades. BMJ Glob Health. 2016; 1(2): e000010. https://doi.org/10.1136/bmjgh-2015-000010 PMID: 28588933
- Hayes R, Floyd S, Schaap A, Shanaube K, Bock P, Sabapathy K, et al. A universal testing and treatment intervention to improve HIV control: one-year results from intervention communities in Zambia in the HPTN 071 (PopART) cluster-randomised trial. PLoS Med. 2017; 14(5):e1002292. <u>https://doi.org/ 10.1371/journal.pmed.1002292</u> PMID: 28464041
- Rosen S, Fox MP, Larson BA, Sow PS, Ehrenkranz PD, Venter F, et al. Accelerating the uptake and timing of antiretroviral therapy initiation in sub-Saharan Africa: an operations research agenda. PLoS Med. 2016; 13(8):e1002106. https://doi.org/10.1371/journal.pmed.1002106 PMID: 27505444